Author's response to reviews

Title: Safety of intramuscular influenza vaccine in patients receiving oral anticoagulation therapy: a single blinded multi-centre randomized controlled clinical trial

Authors:

Josep Casajuana (josep.casajuana@camfic.org)
Begona Iglesias (biglesias@ambitcp.catsalut.net)
Mireia Fabregas (mireiafabregas@gmail.com)
Francesc Fina (ffina@comb.es)
Joan-Antoni Valles (javalles.pbcn@ics.scs.es)
Rosa Aragones (rosaragones@yahoo.es)
Mencia Benitez (mbenitez.pbcn@ics.scs.es)
Edurne Zabaleta (ezabaleta@fundaciojgol.org)

Version: 7 Date: 13 May 2008

Author's response to reviews:

We have revised the manuscript and made the required changes.